Author's response to reviews

Title: Kidney function and multiple hemostatic markers: cross sectional associations in the multi-ethnic study of atherosclerosis

Authors:

Ruth Dubin (ruth.dubin@ucsf.edu)
Mary Cushman (mary.cushman@uvm.edu)
Aaron R. Folsom (folso001@umn.edu)
Linda F. Fried (Linda.Fried@med.va.gov)
Walter Palmas (wp56@columbia.edu)
Carmen A. Peralta (CarmenAlicia.Peralta@ucsf.edu)
Christina Wassel (cwassel@ucsd.edu)
Michael G. Shlipak (Michael.Shlipak@ucsf.edu)

Version: 5 Date: 21 January 2011

Author's response to reviews:

To the editors of BMC,
I have added correct physical address in the affiliations section.
Ruth Dubin